Author's response to reviews

Title: Hospital-acquired invasive pulmonary aspergillosis in patients with hepatic failure

Authors:

Dan Li (lidan@shaphc.org)
Liang Chen (chenliang@shaphc.org)
Xian Ding (dingxian@shaphc.org)
Ran Tao (taoran@shaphc.org)
YongXin Zhang (mzld@hotmail.com)
JieFei Wang (immu_lab@hotmail.com)

Version: 8 Date: 22 July 2008

Author's response to reviews:

Dear editors,

We correct the formatting as requested by the editors.

The details list below:

Page 8, Line 1: The heading is corrected to “Competing Interests”.

Page 8, Line 4: Author’s contributions are formatted into a paragraph rather than a list: All authors contributed to the design of the study. Three authors (Liang Chen, Xian Ding and Ran Tao) collected and analyzed the data, the others made interpretation and final decision when questions arised, two authors (Dan Li and Liang Chen) have been involved in drafting the manuscript.

All authors critically read the manuscript and make sure the submission.

Thanks for your efforts in our work.

With best regards,

Jiefei Wang